Author's response to reviews

Title: Is minimally invasive parathyroid surgery an option for patients with gestational primary hyperparathyroidism?

Authors:

Cino Bendinelli Dr (cino.bendinelli@hnehealth.nsw.gov.au)
Shane Nebauer Mr (shane.nebauer@uon.edu.au)
Tuan Quach Dr (tuan.quach@hnehealth.nsw.gov.au)
Shaun Mcgrath Dr (shaun.mcgrath@hnehealth.nsw.gov.au)
Shamasunder Acharya Dr (shamasunde.acharya@hnehealth.nsw.gov.au)

Version: 3 Date: 14 May 2013

Author's response to reviews:

To The Editors and Reviewers,

We the Authors would like to thank you for your helpful comments on our manuscript. As requested, we have made the following adjustments based upon your feedback:

1. We have revised the grammar, with attention to the correct use of commas in the manuscript.
2. Incorrect capitalisation has been corrected.
3. The titles of some sections have been adjusted.

We hope you find the revised manuscript to be to your satisfaction.

Regards,

The Authors.